Author's response to reviews

Title: Fatal interstitial lung disease associated with oral erlotinib therapy for lung cancer

Authors:

Demosthenes Makris (appollon7@hotmail.com)
Arnaud Scherpereel (a-scherpereel@chru-lille.fr)
Marie Christine Copin (m-copin@chru-lille.fr)
Guillaume Colin (belgutou@yahoo.fr)
Luc Brun (lbrun@chru-lille.fr)
Jean Jacques Lafitte (jjlafitte@chru-lille.fr)
Charles Hugo Marquette (C-MARQUETTE@CHRU-LILLE.FR)

Version: 4  Date: 1 August 2007

Author's response to reviews:

Dear Editor

As you will see in the revised manuscript we made all required changes to the format of our manuscript.

1) Author list - Each author name is on separate line.
2) Short title, Word count were removed.
3) Abstract word count and Key words were removed.
4) Abstract is structured into Background, Case Presentation, Conclusions.
5) We also updated the abstract details on the submission page.
6) Introduction was renamed 'Background'.
7) Case Report was renamed 'Case Presentation'
8) Authors Contribution were changed to Authors' contributions
9) Figure 1 was resubmitted with minimal white space around the image

We hope that this revised version will be considered as suitable for publication

Yours Sincerely,

Arnaud SCHERPEREEL